Author's response to reviews

Title: Prevalence of fibromyalgia in low socioeconomic status population

Authors:

Ana Assumpção (anassumpcao@usp.br)
Alane B Cavalcante (alanebc@gmail.com)
Cristina E Capela (criscapela@terra.com.br)
Juliana F Sauer (juliana.sauer@gmail.com)
Suellen D Chalot (ischalot@uol.com.br)
Carlos AB Pereira (cpereira@ime.usp.br)
Amélia P Marques (pasqual@usp.br)

Version: 5 Date: 31 May 2009

Author's response to reviews: see over
Dear Ms Rikki Graham and Medical Editors of BMC Musculoskeletal Disorders

We would like to thank you for the “pre” acceptance in publishing our manuscript titled: “Prevalence of fibromyalgia in low socioeconomic status population”. We are providing a new, and hopefully the last version of the manuscript. We corrected all points concerned formatting adjustment including the major and minor revisions requested. All the text was reviewed and we did the final correction, especially at references citations and reference section.

Available for any questions, with best wishes

Ana Assumpção and co-authors